Author’s response to reviews

Title: Exclusion and inclusion of parents of hospitalized children in Norway in the period 1877 - 2017

Authors:

Hildegunn Sundal (hildegunn.sundal@himolde.no)

Karin Anna Petersen (Karin.Petersen@uib.no)

Jeanne Boge (Jeanne.Helene.Boge@hvl.no)

Version: 1 Date: 08 Nov 2017

Author’s response to reviews:

We have tried to make the article with the title “Exclusion and inclusion of parents of hospitalized children in Norway in the period 1877-2017” as a research article, with more details in the method and concretized more, and separated the results and the discussion, the way you recommended. The article is sending in two version, one with the correction in red and another without the red writings.

We hope you can advise us if the article is ok, and we would like to know if the article is to be delivered in a different way in addition to sending it to you.

We have received funds to Open access publication in 2017, and the year is at the end. So please let us know if that is possible within 2017.

It is two versions of the article, one with the changes marked in red and one without.